Author's response to reviews

Title: HIV/AIDS status disclosure increases Public Health benefits (Support, behavioral change and HIV prevention) in the long term: A case for an Urban Clinic Kampala, Uganda

Authors:

Lynn M Atuyambe (latuyambe@yahoo.com)
Eric Ssegijja (ssega2001@yahoo.com)
Sarah Ssali (sssalin@yahoo.com)
Christopher Tumwine (xtofa2001@yahoo.com)
Nicolate Nekesa (nicolaten@yahoo.com)
Annet Nannungi (annnannungi@yahoo.com)
Gery Ryan (gery@rand.org)
Wagner Glenn (gwagner@rand.org)

Version: 2 Date: 17 April 2013

Author's response to reviews:

To: Natalie Pafitis, BioMed Central
Executive Editor
BMC Health Services Research
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom
Email: info@biomedcentral.com
Tel: +44 (0) 20 3192 2009
Fax: +44 (0)20 3192 2010

Dear Executive Editor,

MS: 2208593548389330 HIV/AIDS status disclosure increases Public Health benefits (Support, behavioral change and HIV prevention) in the long term: A case for an Urban Clinic Kampala, Uganda

We are very grateful to you for offering us the opportunity to revise our manuscript as per comments and resubmit it to your prestigious Journal BMC Health Services Research. I believe we have been able to address all the
relevant points raised by the 2 reviewers. However we would be willing to consider further revisions if advised so.

Please find attached response to each of the comments.

Thank you for your interest in our work.

Yours faithfully,

Lynn Atuyambe, MPH, Ph.D. –Karolinska Institute/Makerere University
Corresponding author